Author's response to reviews

Title: A pilot study of rivastigmine in the treatment of delirium after stroke; a safe alternative

Authors:

Annemarie AW Oldenbeuving (anne_ olden@hotmail.com)
Paul PLM de Kort (p.d.kort@elisabeth.nl)
Ben BPW Jansen (bjansen@tsz.nl)
Jaap LJ Kappelle (l.kappelle@umcutrecht.nl)
Gerwin G Roks (g.roks@elisabeth.nl)

Version: 5 Date: 18 September 2008

Author's response to reviews: see over
Dear editor,

Below you will find the adjustments we made in response to comments of the editorial production team.

We hope that after these changes our paper is suitable for publication in your journal

Kind regards, also on behalf of the co-authors

Dr. Gerwin Roks

Author list - The list of authors in the manuscript should be written exactly as they are in the submission system. The author list has been changed in the manuscript. Important is that for LJ Kappelle the second initial is his fist name

Affiliations & Each affiliation should be written in full, in the following format: department, institute, city, and country. This may mean repeating institutional addresses if the authors are from different departments within an institution. Can we also ask you to link each author to his or her corresponding affiliation with the use of superscript numbers? Please do not use full capitals for country names, and please do not include full stops in abbreviated country names.

The affiliations have been changed in the manuscript

Manuscript sections - Manuscript sections should include (in the following order): Abstract; Background; Methods; Results; Discussion; Conclusions; Abbreviations (if any); Competing interests; Authors' contributions; Acknowledgements; References; Figure legends (if any); Tables (if any); Description of Additional files (if any).

The sections have been checked and changed if necessary

Conclusions - Please insert a Conclusions section after the Discussion.
This has been done

CONFLICT OF INTEREST -> Competing interests
This has been done

Author¿s contributions -> Authors' contributions
This has been done

Table title and legend ¿ The table title should appear above the table and should be only one sentence long. Please place any other text as a legend (or key), which should appear below the table.
The table title has been moved above the table and the title has been changed in order to fit above the table.

**Figure cropping** - It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image.

This has been done.